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Abstract 

 

To satisfy the energy demand of an increasing population fossil fuel reserves are being 

exploited at an unprecedented rate. The burning of these fossil fuels produces CO2, which is 

the main constituent of greenhouse gases and leads to global warming. Therefore, use of CO2 

as feed for methanol and DME production can be a possible route to close the anthropogenic 

carbon cycle. The reverse water gas shift reaction (RWGS) poses major challenge to the design 

of methanol selective catalyst. Recently oxygen deficient catalytic system like In2O3, ZnO-

ZrO2 and Zn2Ga2O4 have been documented to be very selective for methanol synthesis reaction 

even at high temperature (300°C). The oxygen defective sites are helpful in CO2 activation. 

However, the catalytic activity of these oxygen deficient catalytic system is low due to absence 

of metallic sites for H2 adsorption. Studies reporting interaction between the metallic Cu and 

oxygen deficient supports, and its consequence in the catalytic performance are a few. 

In this regard, a comparative study to investigate the role of oxygen vacancies and basic site 

density in tuning methanol selectivity during CO2 hydrogenation over Cu/CeO2, CuZrO2 and 

Cu/ZnO catalyst was performed. The catalysts were prepared by constant pH co-precipitation 

method and their catalytic performance was evaluated in a fixed bed tubular reactor. The 

methanol selectivity over Cu/CeO2 catalyst reaches 90% at 220°C, 30 bar and 2400 mL gcat
-1 

h-1. The abundance of oxygen defects on the Cu/CeO2 catalyst surface facilities formation of 

unidentate carbonate species, which leads to selective methanol synthesis. The concentration 

of the oxygen vacancies was altered by preparing Cu/CeO2 catalyst with co-precipitation, 

surfactant assisted co-precipitation and sol-gel method. The improved catalytic activity of 

Cu/CeO2 catalyst prepared by sol-gel method is ascribed superior physicochemical properties 

like Cu surface area, small Cu particle size, strong basic site density and oxygen vacancy 

defects. The H2-TPR and XPS results confirms partial reduction of Ce4+ to Ce3+. This reduced 
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CeO2 promotes CO2 adsorption and activation. The methanol space time yield for Cu/CeO2-

SG catalyst reaches 70.8 g kgcat
−1  h-1 at 30 bar, 260 °C, 7200 mL kgcat

−1  h-1 and CO2:H2 ratio 1:3. 

Further as CeO2 supported Cu catalyst was more selective towards methanol synthesis reaction. 

CeO2 was incorporated as promoter into the conventional Cu/ZnO catalyst and for a constant 

Cu wt% (70 wt%), the ratio of Zn/Ce was optimized.  The catalytic activity of Cu/ZnO/CeO2 

catalyst reached maximum for Zn/Ce ratio of 2:1 and was 106.4 g kgcat
−1  h-1 at 30 bar, 250°C, 

3000 mLgcat
-1 h-1 and 1:3 H2/CO2 ratio. The kinetic data was measured within 220°C-310°C 

temperature, 10-30 bar pressure, 3000-7500 mLgcat
-1 h-1 GHSV and 1:1, 1:2, 1:3, 1:4 and 1:5 

CO2:H2 ratio. The dual site LHHW model with formate hydrogenation as the rate determining 

step fitted well to the experimental results.  

Single step DME synthesis poses additional advantage of overcoming the thermodynamic 

barrier for CO2 hydrogenation to methanol. However, excess water produced in the methanol 

dehydration deactivates catalytic performance. Therefore, mildly hydrophobic ZrO2 is used as 

promoter to the Cu/ZnO catalyst. The synergistic effect between Cu, Zn and Zr oxides leads to 

higher DME selectivity and consequently inhibit the side reactions over the bifunctional 

catalyst. The ZnO-ZrO2 interaction leads to formation of oxygen defective sites and improve 

the activity of bifunctional catalyst. The methanol and DME selectivity reache up to 19% and 

60% respectively with 14% CO2 conversion at 30 bar, 260°C, 2400 mLgcat
-1 h-1 and 1:3 CO2/H2 

ratio. Altogether the tuning metal oxide interactions and oxygen vacancy defects is the keys to 

enhance methanol yield by CO2 hydrogenation.  
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साराांश  

एक बढ़ती हुई आबादी की ऊर्ाा माांग को पूरा करने के लिए र्ीवाश्म ईांधन का अभूतपूवा दर से दोहन 

लकया र्ा रहा है। इन र्ीवाश्म ईांधनोां के र्िने से CO2 पैदा होती है, र्ो ग्रीनहाउस गैसोां का मुख्य घटक 

है और ग्लोबि वालमिंग की ओर िे र्ाती है। इसलिए, मेथनॉि और डीएमई उत्पादन के लिए फीड के 

रूप में CO2 का उपयोग मानवर्लनत काबान चक्र को बांद करने का एक सांभालवत मागा हो सकता है। 

ररवसा वॉटर गैस लिफ्ट ररएक्शन (RWGS) मेथनॉि चयनात्मक उते्प्ररक के लडर्ाइन के लिए बडी चुनौती 

है। हाि ही में ऑक्सीर्न अभावपूण उते्प्ररक रै्से In2O3, ZnO-ZrO2 और Zn2Ga2O4 को उच्च तापमान 

(300 लडग्री सेल्सियस) पर भी मेथनॉि सांशे्लषण प्रलतलक्रया के लिए बहुत चयनात्मक होने के लिए प्रिेल्सित 

लकया गया है। ऑक्सीर्न दोषपूणा स्थि CO2 सलक्रयण में सहायक होते हैं। हािाांलक, H2 अविोषण के 

लिए धातु स्थिोां की अनुपल्सस्थलत के कारण इन ऑक्सीर्न की कमी वािे उते्प्ररको की उते्प्ररकता 

गलतलवलध कम है। धातुयुक्त सहायकोां और ऑक्सीर्न दोषपूणा समथान के बीच सांवेदनिीिता में अवसरोां 

की ररपोटा करने वािे अध्ययनोां पररणाम बहूत कम हैं। 

इस सांबांध में, Cu/CeO2, CuZrO2 और Cu/ZnO उते्प्ररक पर CO2 हाइडर ोर्नीकरण के दौरान मेथनॉि 

चयनात्मकता समायोर्न में ऑक्सीर्न ररल्सक्तयोां और बुलनयादी साइट घनत्व की भूलमका की र्ाांच करने 

के लिए एक तुिनात्मक अध्ययन लकया गया था। उते्प्ररक लनरांतर पीएच सह-वषाा लवलध द्वारा तैयार लकए 

गए थे और उनके उते्प्ररक प्रदिान का मूल्ाांकन लिक्स्ड बेड टू्यबिर ररएक्टर में लकया गया था। 

Cu/CeO2 उते्प्ररक पर मेथनॉि चयनात्मकता 220°C, 30 बार और 2400 mL gcat-1 h-1 पर 90% तक 

पहुांच र्ाती है। Cu/CeO2 उते्प्ररक सतह सुलवधाओां पर ऑक्सीर्न दोषोां की प्रचुरता एकरूप काबोनेट 

प्रर्ालतयोां का लनमााण करती है, र्ो चयनात्मक मेथनॉि सांशे्लषण की ओर िे र्ाती है। सह-वषाा, सिैक्टेंट 

सहायक सह-वषाा और सोि-रे्ि लवलध के साथ Cu/CeO2 उते्प्ररक तैयार करके ऑक्सीर्न ररल्सक्तयोां 

की एकाग्रता को बदि लदया गया था। सोि-रे्ि लवलध द्वारा तैयार Cu/CeO2 उते्प्ररक की बेहतर उते्प्ररक 
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गलतलवलध को Cu सतह के्षत्र, छोटे Cu कण आकार, मर्बूत बुलनयादी साइट घनत्व और ऑक्सीर्न ररल्सक्त 

दोष रै्से बेहतर भौलतक-रासायलनक गुण लदए गए हैं। H2-TPR और XPS पररणाम Ce4+ के Ce3+ में 

आांलिक कमी की पुलि करते हैं। CO2 लवज्ञापन और सलक्रय करने को बढ़ावा लदया है। Cu/CeO2-SG 

उते्प्ररक के लिए मेथनॉि से्पस टाइम यील्ड 70.8 g kgcat
-1 h-1 30 बार, 260 °C, 7200 mL kgcat

-1 h-

1 और CO2:H2 अनुपात 1: 3. इसके अिावा CeO2 समलथात Cu उते्प्ररक मेथनॉि सांशे्लषण प्रलतलक्रया 

के प्रलत अलधक चयनात्मक था। CeO2 को पारांपररक Cu/ZnO उते्प्ररक में प्रमोटर के रूप में िालमि 

लकया गया था और ल्सस्थर Cu wt% (70 wt%) के लिए, Zn/Ce के अनुपात को अनुकूलित लकया गया 

था। Cu/ZnO/CeO2 उते्प्ररक की उते्प्ररक गलतलवलध 2:1 के Zn/Ce अनुपात के लिए अलधकतम पहुांच 

गई और 30 बार, 250°C, 3000 mLgcat
-1 h-1 पर 106.4 g kgcat

-1 h-1 थी और 1:3 H2/CO2 अनुपात। 

काइनेलटक डेटा को 220°C-310°C तापमान, 10-30 बार पे्रिर, 3000-7500 mLgcat
-1 h-1 GHSV 

और 1:1, 1:2, 1:3, 1:4 और 1: 5 CO2:H2 अनुपात  के भीतर मापा गया था। दर लनधााररत करने वािे 

चरण के रूप में फॉमेट हाइडर ोर्नीकरण के साथ दोहरी साइट LHHW मॉडि प्रयोगात्मक पररणामोां के 

लिए अच्छी तरह से लिट है। 

लसांगि से्टप डीएमई सांशे्लषण से मेथनॉि में CO2 हाइडर ोर्नीकरण के लिए उष्मागलतकी  बाधा पर काबू 

पाने का अलतररक्त िाभ लमिता है। हािाांलक, मेथनॉि लनर्ािीकरण में उत्पालदत अलतररक्त पानी उते्प्ररक 

प्रदिान को लनल्सिय कर देता है। इसलिए, हले्क र्िलवरोधी ZrO2 का उपयोग Cu/ZnO उते्प्ररक के 

प्रवताक के रूप में लकया र्ाता है। Cu, Zn और Zr ऑक्साइड के बीच सहलक्रयात्मक प्रभाव उच्च DME 

चयनात्मकता की ओर िे र्ाता है और पररणामस्वरूप लद्वभालर्त उते्प्ररक पर पार्श्ा प्रलतलक्रयाओां को 

रोकता है। ZnO-ZrO2 अन्योन्य लक्रया ऑक्सीर्न दोषपूणा साइटोां के गठन की ओर िे र्ाती है और लद्व-

कायाात्मक उते्प्ररक की गलतलवलध में सुधार करती है। 30 बार, 260°C, 2400 mLgcat
-1 h-1 और 1:3 

CO2/H2 अनुपात में 14% CO2 रूपाांतरण के साथ मेथनॉि और DME चयनात्मकता क्रमिः  19% और 



   

 

viii 

 

60% तक पहुांच र्ाती है। कुि लमिाकर टू्यलनांग मेटि ऑक्साइड इांटरैक्शन और ऑक्सीर्न ररल्सक्त दोष 

CO2 हाइडर ोर्नीकरण द्वारा मेथनॉि उपर् बढ़ाने की कुां र्ी है। 
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